Author's response to reviews

Title: Role of hemoclip in the management of acute bleeding from gastric stromal tumor: A Case Report and review of the literature

Authors:

Mouen A Khashab (mkhashab@iupui.edu)
Harvery M Cramer (hcramer@iupui.edu)
Suthat Liangpunsakul (sliangpu@iupui.edu)

Version: 5 Date: 25 August 2007

Author's response to reviews:

- corrected some spelling and grammatical mistakes
- deleted figure 1
- proper way of writing the medication name addressed